

Adverse Drug Reactions in Patients with CKD

Solène Laville, Valérie Gras-Champel, Julien Moragny, Marie Metzger, Christian Jacquelinet, Christian Combe, Denis Fouque, Maurice Laville, Luc Frimat, Bruce Robinson, et al.

► To cite this version:

Solène Laville, Valérie Gras-Champel, Julien Moragny, Marie Metzger, Christian Jacquelinet, et al.. Adverse Drug Reactions in Patients with CKD. Clinical Journal of the American Society of Nephrology, 2020, 15 (8), pp.1090-1102. 10.2215/cjn.01030120 . hal-02958497

HAL Id: hal-02958497 https://hal.science/hal-02958497v1

Submitted on 5 Oct 2020

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Adverse Drug Reactions in Patients with Chronic Kidney Disease

Authors: Solène M Laville¹, Valérie Gras-Champel², Julien Moragny², Marie Metzger¹, Christian Jacquelinet^{1,3}, Christian Combe^{4,5}, Denis Fouque⁶, Maurice Laville⁶, Luc Frimat^{7,8}, Bruce M. Robinson⁹, Bénédicte Stengel¹, Ziad A Massy^{1,10}, Sophie Liabeuf^{2,11} on behalf of the Chronic Kidney Disease-Renal Epidemiology and Information Network (CKD-REIN) Study Group

Authors' institutions:

¹ Centre for Research in Epidemiology and Population Health (CESP), INSERM UMRS 1018, Université Paris-Saclay, Université Versailles Saint Quentin, F-94807 Villejuif, France

² Department of Clinical Pharmacology, Amiens University Hospital, F-80000 Amiens, France
 ³ Biomedecine Agency, F-93210 Saint Denis La Plaine, France

⁴ Service de Néphrologie Transplantation Dialyse Aphérèse, Centre Hospitalier Universitaire de Bordeaux, Bordeaux, France

⁵ INSERM, U1026, Univ Bordeaux Segalen, Bordeaux, France

⁶Nephrology Dept, Centre Hospitalier Lyon Sud, Université de Lyon, Carmen, F-69495 Pierre-Bénite, France

⁷ Nephrology Department, CHRU de Nancy, F-54000 Vandoeuvre-lès-Nancy, France

⁸ Lorraine University, APEMAC, F-54000 Vandoeuvre-lès-Nancy, France

⁹ Arbor Research Collaborative for Health, Ann Arbor MI, USA

¹⁰ Department of Nephrology, Ambroise Paré University Hospital, APHP, F-92104 Boulogne-Billancourt/Paris, France

¹¹ MP3CV Laboratory, EA7517, University of Picardie Jules Verne, Amiens, F-80000 Amiens, France

Running title: Adverse drug reactions in CKD

Corresponding author:

Bénédicte Stengel Director, Inserm U1018, Team 5 UPS-UVSQ, CESP, Centre for Epidemiology and Population Health EpReC Team, Renal and Cardiovascular Epidemiology 16, avenue P. Vaillant Couturier F-94 807 Villejuif cedex France Phone: + 33-145-595-039 E-mail: benedicte.stengel@inserm.fr

Keywords: Pharmacoepidemiology, renin-angiotensin system inhibitors, antithrombotic agents, diuretics, chronic kidney disease, adverse drug reaction

SIGNIFICANCE STATEMENT

1. What was previously known about the specific topic of the manuscript?

Patients with CKD have complex presentations, use multiple medications, often receive inappropriate prescriptions, and are at high risk for adverse drug reactions (ADRs). Investigations of ADRs in outpatients with CKD have been sparse.

2. What were the most important findings?

Patients with CKD have a high incidence of ADR, even when they receive care from nephrologists. Some cases are serious (20%), especially those attributable to antithrombotic agents. Most ADRs resulted in discontinuation of the drug to which they were attributed (71%). As kidney function deteriorated during CKD, ADR incidence rose.

<u>3. How does the new information advance a new understanding of the kidney and its</u> <u>diseases?</u>

Many serious ADRs in patients with moderate or advanced CKD may be preventable. Greater awareness of eGFR by prescribers and pharmacists and enhanced patient education are key elements for reducing ADRs in this population at especially high risk.

ABSTRACT

Background: Little is known about the burden of adverse drug reactions (ADRs) in chronic kidney disease (CKD). We estimated the incidence of overall and serious ADRs and assessed the probability of causation, preventability, and factors associated with ADRs in patients seen by nephrologists.

Methods: The CKD-REIN cohort included 3033 outpatients (65% men) with CKD and estimated GFR (eGFR) <60 mL/min/1.73m², with follow-up for two years. ADRs were identified from hospitalization reports, medical records, and participant interviews, and finally assessed for causality, preventability, and immediate therapeutic management by experts in pharmacology.

Results: Median (interquartile range) age was 69 (60–76); 55% had eGFR≥30 and 45% <30 mL/min/1.73m². Participants were prescribed a median (range) of 8 (5–10) drugs. Over 2 years, 536 patients had 751 ADRs, 150 (in 125 participants) classified as serious, for rates of 14.4 (95% confidence interval, 12.6–16.5) and 2.7 (1.7–4.3) per 100 person-years, respectively. Among the serious ADRs, 32% were considered preventable or potentially preventable; 16 caused death, directly or indirectly. Renin-angiotensin system inhibitors (15%), antithrombotic agents (14%), and diuretics (10%) were the drugs to which the most ADRs were imputed, but antithrombotic agents caused 34% of serious ADRs. The drug was discontinued in 71% of cases, at least temporarily. Adjusted hazard ratios for serious ADR were significantly higher in patients with eGFR <30 $vs \ge$ 30 mL/min/1.73m², 1.8 (1.3-2.6), in those prescribed >10 *versus* <5 medications, 2.4 (1.1-5.2), or with poor *versus* good adherence, 1.6 (1.4-2.4).

Conclusion: ADRs are common and sometimes serious in patients with CKD. Many serious ADRs may be preventable. Some specific pharmacological classes, particularly antithrombotic agents, are at risk of serious ADRs.

INTRODUCTION

Despite the highly regulated process of drug marketing authorization, no medicine is completely safe. Adverse drug reactions (ADRs) are relatively common; they cause 2–7% of overall hospitalizations.(1–4) Two French studies (2,5) have reported that 3.2% to 3.6% of hospital admissions are related to ADRs. The concept of ADR has evolved over time and today includes any harmful and unwanted reaction to a drug, occurring at doses normally used in humans, or resulting from drug misuse or error, or from accidental or willful overdose.(6) Most available studies have focused on ADR-related hospitalizations (with the ADR either the cause of admission (2,3,5,7–10) or occurring during hospitalization,(11,12) or both (1,13,14)); very few have examined ADRs in outpatient settings.(4) Although clinicians recognize ADRs as a major problem in CKD patients, few studies have investigated the incidence of and factors associated with ADRs in this population.

Drug-related nephrotoxicity is frequent and well documented.(15) But the kidney also plays an important role in the clearance of many drugs and toxic metabolites that can cause ADR due to their accumulation as kidney function declines. Despite impaired pharmacokinetics and pharmacodynamics,(16–18) patients with CKD use multiple medicines and are often exposed to some that are inappropriately prescribed.(19) Until now, studies in CKD patients have been based on self-reported ADRs,(20) or have concerned specific drugs during clinical trials,(21,22) or have been restricted to ADRs during hospitalization,(23) or to specific types of ADRs.(24) None reported therapeutic management after the ADR. No comprehensive evaluation exists of the incidence, probability of causation, and preventability of ADRs in both inpatients and outpatients belonging to this population.

The primary objective of this study was to estimate the incidence rates of overall and serious ADRs according to eGFR, in patients with moderate or advanced CKD treated by a

nephrologist. Secondary objectives aimed at assessing their causation, preventability, associated factors, and immediate therapeutic management.

METHODS

Study design and participants

Chronic Kidney Disease-Renal Epidemiology and Information Network (CKD-REIN) is a prospective cohort study conducted in 40 nationally representative nephrology outpatient facilities in France. Eligible patients were at least 18 years of age, had a confirmed diagnosis of moderate or advanced CKD, an eGFR <60 mL/min per 1.73 m², were not on dialysis, and had not been transplanted. From July 2013 to March 2016, CKD-REIN included 3033 patients. Details of the study protocol and flow chart have been published elsewhere.(25) The institutional review board of the French National Institute of Health and Medical Research (INSERM; reference: IRB00003888) approved the protocol, and the study was registered at ClinicalTrials.gov (NCT03381950).

Information

Data were collected at baseline and then annually by trained clinical research associates (CRAs) from participants interviews and medical records from the nephrology centers that included them. All of them contain patient histories, hospitalizations reports, imaging and laboratory data from every ward of the hospital/clinic, but they are not standardized and may differ between these centers. Data collected included sociodemographic characteristics and a history of hypertension, diabetes, cardiovascular disease, dyslipidemia, or acute kidney injury (AKI), as defined previously.(25) Medication adherence was assessed with the Girerd score,(26) calculated by asking 6 questions that explore the primary determinants of adherence to chronic medication (its timing, remembering to take it, and remembering to renew the prescription).

Serum creatinine, albumin, and hemoglobin were measured, as was urinary albumin or protein. We used the CKD-EPI equation to estimate GFR.(27) Participants were asked to bring to their inclusion appointment all their current drug prescriptions for the previous three months (regardless of the prescribing physician) and all prescriptions for the year to each annual followup appointment. In France, all prescriptions are reimbursed similarly, except for medications determined by the Ministry of Health to have only moderate or insufficient medical benefits. Some differences in reimbursement also exist for patients with chronic expensive diseases and for patients who have not purchased supplementary health insurance, but none of these differences affect the recording and processing of prescriptions. Accordingly, drug prescriptions were continuously recorded from 3 months preceding inclusion through the end of follow up. We used the international Anatomical Therapeutic and Chemical (ATC) thesaurus (28) to code treatments and recorded their start and discontinuation dates (with causes, if any). Kidney failure events, defined as dialysis start or preemptive transplantation, and deaths were reported by the participants or their families, or were identified from medical records or record linkage with the national kidney failure registry.(29)

Identification and validation of adverse drug reactions

An ADR is defined as "an appreciably harmful or unpleasant reaction, resulting from an intervention related to the use of a medicinal product, which predicts hazard from future administration and warrants prevention or specific treatment, or alteration of the dosage regimen, or withdrawal of the product".(6) An ADR is considered serious when the patient outcome is death (or a life-threatening situation), hospitalization (initial or prolonged), disability or permanent damage, or another important medical event.(30)

We collected ADRs over a two-year follow-up via an electronic form designed specifically to include information critical for this study. We used several sources to identify ADRs: (1) medical records (examined by CRAs) (2) participant interviews by CRAs, and (3) hospital

reports (Figure 1). Hospitalizations were identified from (1) electronic medical records, (2) nephrology records, and (3) participant interviews. For each hospitalization, we obtained a report to confirm the period and the cause. Causes of hospitalization were coded by a physician according to the 10th revision of the International Classification of Diseases (ICD-10). Every drug prescribed to the patient at the time of each ADR was recorded. Each identified ADR was reviewed by two pharmacists (SL and SML), who evaluated the potential causation of reported drug-related ADRs and coded the types of effect according to the medical dictionary for regulatory activities (MedDRA® Dictionary), the severity of the ADR (nonserious or serious), the drug suspected of responsibility for the ADR, its dosage, and immediate medication management: discontinuation of the product, dose adaptation, or no change. If the ADR was considered serious, a larger committee of expert pharmacologists (VGC, JM, SL, and SML) from the Amiens pharmacovigilance center further evaluated the potential causal relation of each drug (prescribed at the time of the ADR) and the preventability of the ADR.

Assessment of the causes and preventability of serious adverse drug reactions

We applied the Bégaud imputability method (31) (Supplementary material 1A), which is the official procedure used in French pharmacovigilance centers to report serious ADRs to the French drug authorities. This algorithmic method attributes an intrinsic score based on chronological and semiological criteria. The cause-and-effect relationship is assessed independently for each drug taken by the patient before the occurrence of the event and is not influenced by the extent of imputability to other drugs. This method allowed us to identify the drug most responsible for serious ADRs, i.e., that with the highest intrinsic score. During the ADR validation process, we globally evaluated all of the risk factors for ADRs, including a review of all drugs prescribed at the time of the ADR and evaluation of potential pharmacokinetic/pharmacodynamic interactions.

In addition, we used the Naranjo 10-question algorithm (Supplementary material 1B) to confirm the causal relation of each reported serious ADR by determining the probability that an ADR is actually due to a drug rather than to any other factor.(32) This analysis considers only ADRs categorized as definite, probable, or possible with both the Bégaud and Naranjo methods.

The preventability of ADRs was assessed with a seven-item ADR preventability scale (33) (Supplementary material 1C) that classified ADRs in four categories: "preventable", "potentially preventable", "not assessable", "not preventable". When items related to adherence to recommendations and the patient's need for the prescription were uncertain, the expert committee rated preventability as "not assessable".

Both outpatient and inpatient medical records were used to assess causality and preventability.

Statistical analyses

Baseline characteristics were described for all participants and by subgroup according to baseline eGFR ($< or \ge 30 \text{ mL/min per } 1.73 \text{ m}^2$). ADRs were also described according to baseline eGFR. Results were expressed as means \pm standard deviations (SD), medians (interquartile range), or numbers (percentages). The Fisher exact, Student *t*, or Chi-squared tests were used to compare categorical variables. A sensitivity analysis was conducted by using the last eGFR preceding the ADR to describe it.

Crude incidence rates and 95% confidence intervals (95% CIs) of ADRs and serious ADRs per 100 person-years were estimated by Poisson regression for the overall population and by subgroups according to baseline eGFR; they were corrected for overdispersion by using the quasi-likelihood approach.

We used cause-specific Cox proportional hazard models to investigate patient clinical characteristics associated with the ADR risk. Data were censored at the end of the two-year follow-up, the last patient visit, death, or kidney failure, whichever came first (competing

events). Variables from Table 1, preselected by a literature review, were analyzed in a crude model. Variables with a *P* value greater than 0.10 in the crude model were excluded from the multivariable analyses. Age and sex were forced into the final model, that is, included because considered necessary for the model's validity.

Because of the possibility of multiple ADRs per patient during the follow-up, we performed a sensitivity analysis, using the Prentice, Williams, and Peterson gap-time recurrent event time-to-event analysis, with sandwich variance estimators, to determine the factors associated with ADRs.(34,35)

To deal with the missing data (Supplementary material 2), multiple imputations were performed (fully conditional specification method,(36) 10 iterations, 10 datasets) including all patient characteristics from Table 1, the total number of ADRs per patient, and the number of serious ADRs per patient. Data patterns suggest that the assumption that data were missing at random was plausible. Cox model regression coefficients were estimated separately in each imputed dataset and combined according to Rubin's rules.

Statistical analyses were performed with SAS® software (version 9.4; SAS Institute, Cary, NC) and R software (version 3.5.0, Foundation for Statistical Computing, Vienna, Austria).

RESULTS

Baseline characteristics

Participants were predominantly men; 43% had diabetes, 53% cardiovascular disease, and 91% hypertension. (Table 1) Those with eGFR < 30 mL/min per 1.73 m² at baseline were older and had less education than those with higher eGFR. They also had anemia and a history of AKI more often, and were prescribed more medications.

Incidence and description of adverse drug reactions

Over the two-year follow-up, 751 ADRs were reported in 536 (18%) of 3033 participants; 150 ADRs in 125 participants (4%) were classified as serious, i.e., 14.4 (95% CI, 12.6-16.5) and 2.7 (95% CI, 1.7-4.3) per 100 person-years, respectively. Both were nearly twice as high in participants with eGFR <30 versus \geq 30 mL/min per 1.73m² (Table 3, Table 4, and Figure 2). Five percent of participants had more than one ADR. Sixty percent of ADRs were reported only in the medical record or hospitalization report, 20% in both the medical record and participant interview, and 20% only in the participant interview.

Among the serious ADRs, 145 were associated with hospitalization: 65% as its cause and 35% as its consequence (Figure 3). Sixteen deaths resulted from an ADR, directly or indirectly, as did five life-threatening events. Five participants with serious ADRs (three medically important and two resulting in permanent disability) were not hospitalized.

Renal and urinary disorders were the most frequent type of ADR, particularly AKI, followed by gastrointestinal (mostly diarrhea), and musculoskeletal and connective tissue disorders (Table 2 and supplementary material 3). Renal disorders and hemorrhages or bleeding accounted for two thirds of the 150 serious ADRs (Figure 3). Of the 16 deaths linked to an ADR, 11 were related to hemorrhages, which also accounted for 40% of the serious ADRs in participants with baseline eGFR < 30 mL/min per 1.73m², and 19% in those with eGFR \geq 30 (*P*=0.009) (Figure 3). Using the last eGFR before the ADR did not significantly change the proportions reported in Table 2 and Figure 3 (Supplementary materials 4 and 5).

Renin-angiotensin system (RAS) inhibitors, antithrombotic agents, and diuretics were the medications most frequently responsible for both nonserious and serious ADRs (Figure 4 and supplementary material 6). Antithrombotic agents were responsible for 34% of the 150 serious ADRs: 34 were due to vitamin K antagonists, 9 to heparin, 6 to platelet aggregation inhibitors, and 2 to direct factor Xa inhibitors. Three pharmacodynamic drug interactions were identified:

two between vitamin K antagonist and heparins, and one between vitamin K antagonist and an antibiotic.

Factors associated with adverse drug reactions and serious adverse drug reactions

Participants with a baseline eGFR <30 mL/min per $1.73m^2$ had a risk 1.6 times higher of an ADR than those with an eGFR \geq 30, after adjustment for other associated variables (Table 3). The risk of ADR also significantly increased with the participant's baseline number of prescribed drugs, history of cardiovascular disease and of AKI, and poor treatment adherence. Women were at higher risk of an ADR than men, but not of a serious ADR. There was no significant association with age.

Hazard ratios for serious ADRs were significantly higher in participants with eGFR < 30 compared to \geq 30 mL/min per 1.73m², as well as in those prescribed >10 compared with <5 medications and in participants with poor adherence (Table 4). Age and sex were not associated with a higher risk of serious ADR.

A sensitivity analysis confirmed that the factors mentioned above were associated with ADRs (Supplementary material 7A). It also showed that anemia was significantly associated with the risk of serious ADRs but that poor adherence no longer was (Supplementary material 7B).

Immediate management of adverse drug reactions and preventability

After an ADR, the drug considered responsible was discontinued in 71% of cases (e.g. in 78% of statin-linked ADRs), at least temporarily, and the dose was adjusted in 14% of cases; no change was made in the prescription in 11% (4% missing data). When the ADR was serious, 83% of the drugs blamed were discontinued at least temporarily right after the event.

The Olivier ADR preventability scale (33) allowed us to classify the 150 serious ADRs as preventable in 13% of cases (n=19) and potentially preventable in 19% (n=28). A quarter of the

preventable ADRs were associated with participant self-medication. Overall, 37% of ADRs were inevitable and 31% not assessable (Supplementary material 6).

DISCUSSION

This study presents a global descriptive view of the magnitude and diversity of ADRs in a well-phenotyped CKD population. The central message here is that ADRs are common, often serious, and potentially preventable in CKD patients and that these patients are vulnerable and their treatment is complex. It shows that three drug classes among those most prescribed in this population are responsible for almost 40% of ADRs, including RAS inhibitors, antithrombotic agents, and diuretics. The study especially points out the severity of ADRs caused by antithrombotic agents, to which one third of the serious events were imputed. In addition, we identify some care and patient characteristics that increase ADR risk; these include an eGFR (< 30 mL/min/1.73m²), a higher number of prescribed drugs, and poor adherence to medications. Importantly, a significant proportion of these ADRs may be preventable.

Our findings are difficult to compare with those of other studies, which differ from ours in several ways. Most of them assessed ADR incidence at hospital admission (1–3,5,7–10,13) or during hospitalization,(1,11–13,23) and used highly heterogeneous study settings, definitions of ADRs, and methods for ascertaining ADRs (spontaneous report, intensive chart review, or both). In this study, we evaluated ADR incidence in nephrology outpatients, with and without hospitalization; they were identified from an extensive review of medical records, hospitalization reports, and participant interviews. We compared rates according to eGFR and showed that ADR incidence increased when eGFR was lower than 30 mL/min/1.73m². Overall, renal, urinary, gastrointestinal, musculoskeletal, and connective tissue disorders were the most commonly reported ADRs in our study, but renal disorders and bleeding largely predominated among the serious ADRs. Several studies have reported similar results in hospitalized patients with unknown CKD status.(5,8–10) The known high susceptibility of patients with CKD for AKI explains the high frequency of renal and urinary ADRs. Cardiovascular medicines stand out among the most common suspected drugs in several studies of hospitalized patients with unknown CKD status.(2–5,9,10) Similarly, we found that renin-angiotensin system inhibitors, antithrombotic agents, and diuretics were the pharmacological classes to which ADRs were most commonly imputed in our study. The high prevalence of CKD-related cardiovascular complications explains the high use of cardiovascular drugs in patients with CKD(37) despite their high risk of ADR due to the combined effect of low eGFR, hemorrhagic risk, and electrolyte disturbance. The principal drugs suspected of causing ADRs were usually not directly nephrotoxic, and most ADRs resulted from reduced renal clearance. This conclusion has important clinical implications, notably regarding drug prescriptions for CKD patients and the need to focus on regularly reassessing use or dose according to eGFR (especially when it drops below 30 mL/min/1.73m²), as well as on potential nephrotoxic agents. However, the clinical benefits of some of these drugs have been demonstrated by a high level of evidence. For instance, a moderate increase (of 20% to 30%) in creatinine can be expected with RAS inhibitors; nephrologists may find this an acceptable trade-off in view of these drugs' protective nature in the long term and their ability to slow CKD progression. This moderate increase is well below the KDIGO definition used for drug-related AKI events in our study, which is based on a rise in creatinine of at least 50%.(25)

Patients seen by nephrologists require the most complex care, because of their multiple comorbidities and the complications associated with decreased kidney function.(38) These result in the use of multiple medications, as shown here: CKD-REIN participants were prescribed a median of eight different drugs daily. Moreover, the number of prescribed drugs increases as CKD progresses.(19,39) High rates of ADRs in patients with CKD add to the complexity of their care. As we showed, treatments often need to be stopped, at least temporarily, or dosage adjusted because of ADRs, which may affect therapeutic management

and ultimately reduce the likelihood of slowing CKD progression and decreasing its complications. Lipid-lowering agents are a good example of treatments that are strongly recommended in patients with CKD due to their high cardiovascular risk, but are often discontinued because of contractures, cramps, or myalgia (stopped in 78% of statin-linked ADRs). Schneider et al. (40) noted the underprescription of statins in CKD patients, which may be partly explained by ADRs; a similar underprescription rate exists in the CKD-REIN cohort.(41) The challenge for physicians is to assess the benefit/risk ratio between treating a new complication and adding a new drug. Increased awareness by the medical community of this difficulty and of the necessity to reassess this benefit/risk ratio regularly, especially when eGFR decreases, is essential. Pharmacists too must play a role, given that most ADRs occur in outpatients. Finally, the importance of patient education in terms of drug use must be enhanced, as one quarter of preventable serious ADRs were due to participant misuse.

The major factors identified in this study associated with ADRs overall and with serious ADRs are the number of prescribed drugs, cardiovascular disease, poor adherence to treatment, and eGFR. Reports have regularly shown that the number of drugs is a risk factor in hospitalized patients with unknown CKD status.(8,9,14,23) Besides the increased risk of ADR, polypharmacy is also associated with deleterious health outcomes in elderly patients.(42) Poor adherence to medication is associated with higher ADR risk, possibly reflecting patients' misuse of drugs. However, the healthy user effect cannot be ruled out.(43) Two other studies have also described cardiovascular disease as a risk factor for ADR.(7,23) Although numerous studies show women to be at higher risk of serious ADR in populations with an unknown CKD status,(2,9,10,14) this was not the case in our CKD population. Age was not associated with the risk of either ADR or serious ADR, consistent with previous reports.(8,14,23) The risk of ADR tended to be slightly lower in older patients in the multivariable analyses, although not significantly; this finding may be related to their lower number of inappropriate

prescriptions.(19) This may indicate more careful prescriptions by physicians for elderly patients. Declining kidney function seems to be an important risk factor for CKD patients with eGFR < compared with \geq 30 mL/min per 1.73 m², in line with the results reported by Sharif-Askari et al. in hospitalized CKD patients.(23) Corsonello et al. found a similar association between ADR risk and declining kidney function in elderly hospitalized patients.(44) This higher rate of ADR at eGFR < 30 mL/min/1.73m² appears related mainly to bleeding events due to antithrombotic agents, mostly vitamin K antagonists. This highlights the importance of reassessing prescriptions regularly as kidney function declines, especially for drugs such as vitamin K antagonists. Although these medications are metabolized by the liver, excreted in an inactive form in stool and urine,(45) and thus not renally eliminated, CKD can impair their disposition.(18) Specifically, it can cause the accumulation of uremic toxins in blood and organs with deleterious effects,(46) for example, indoxyl sulfate, which impairs platelet activity.(47)

Major strengths of this study include its large sample size of patients with confirmed CKD diagnoses, recruited from a representative sample of nephrology outpatient facilities. The high sensitivity and specificity of our process for identifying and grading ADRs is also unique. Indeed, a major limitation in ADR research is the lack of reliable data about the true burden of these reactions, related to their reporting process; the traditional ascertainment method, through the spontaneous reporting system in pharmacovigilance, results in their vast underreporting. A previous study showed that health professionals spontaneously report only 6% of serious ADRs.(48) Our study used different sources to capture these events, especially serious ADRs, through hospitalization reports, medical records, and participant interviews. Furthermore, all cases were reviewed by pharmacists, and serious cases evaluated by a committee of experts. Finally, the use of standard international coding systems for both drug classes and ADRs, the availability of start and end dates for each treatment, and the reasons for treatment discontinuation contribute to the quality and novelty of our findings.

Our study also has limitations. Despite our sensitive method for identifying ADRs, their number may still be underestimated, mainly for those not hospitalized during follow-up: these may not have been reported by physicians in medical records or by participants either to physicians or during interviews, due to memory bias or because an ADR is so well-known that it tends to be poorly reported in hospitalization reports and medical records. However, participants were probably most likely to report ADRs with the most negative impacts on their health and well-being. We may not have captured all hospitalizations and may also have missed drug effects that have never been reported as a potential ADR. Other sources of ADRs might exist, but were not available to us. Several decision algorithms for causality assessment in ADRs exist.(49) However, none of them have been accepted as a gold standard and comparing them could lead to discrepancies in results.(50) These algorithms do not replace medical diagnosis.

Another limitation of this study is the heterogeneous nature of the outcome (ADR) used in the multivariable analyses. Although these analyses allowed us to identify some specific and potentially modifiable risk factors, our basic aim here was to describe the general characteristics of patients at high risk for ADR. We will further explore risk factors associated with ADRs more specifically for a number of drug classes and outcomes.

Finally, our study is generalizable to patients with CKD seen by nephrologists, but not to all patients with an eGFR <60 mL/min per $1.73m^2$ in the general population. However, because nephrologists probably handle drugs more carefully, the ADR incidence would likely have been even higher in the broader population of patients not seen by nephrologists.

CONCLUSION

The burden of ADRs is high in patients with moderate to advanced CKD, and incidence was higher when CKD was severe. Our results highlight the major risk of specific pharmacological

classes, particularly antithrombotic agents, which must be used cautiously in CKD patients, especially at low eGFRs. Greater awareness by the medical community of the importance of eGFR level in prescribing medications, increased involvement of pharmacists in systematically verifying eGFR for patient prescriptions, and enhanced patient education are key elements for preventing ADRs in this population at high risk. The impact of ADRs on health resources and patients' quality of life requires further evaluation.

Author contributions:

S.M.L., S.L., Z.A.M., B.S., and M.M. designed the present project. S.M.L., S.L., V.G-C., and J.M. evaluated the serious adverse drug reactions. S.M.L., S.L., and M.M. analyzed the data. S.M.L., S.L., V.G-C., J.M., Z.A.M., B.S., and M.M. contributed to the interpretation of the results. S.M.L., S.L., Z.A.M., and B.S. wrote the first draft of the article; S.M.L., V.G-C., J.M., M.M., C.J., C.C., D.F., M.L., L.F., B.M.R., B.S., Z.A.M., and S.L. provided critical feedback, helped shape the research, analysis and the final draft of the manuscript, and approved the version to be published; B.S. was the principal investigator.

Disclosures

S.M.L., M.M., V.G-C., J.M., C.J., S.L. have nothing to declare. Z.A.M. reports grants for CKD-REIN and other research projects from Amgen, Baxter, Fresenius Medical Care, GlaxoSmithKline, Merck Sharp and Dohme-Chibret, Sanofi-Genzyme, Lilly, Otsuka, and the French government, as well as fees and grants to charities from Amgen, Daichii, and Sanofi-Genzyme. These sources of funding are not necessarily related to the content of the present manuscript. B.S. reports grants for CKD-REIN from Amgen, Baxter, Fresenius Medical Care, GlaxoSmithKline, Merck Sharp and Dohme-Chibret, Sanofi-Genzyme, Lilly, Otsuka, and Vifor Fresenius, as well as speaker honoraria at the French Society of Diabetology from Lilly, and at the French-speaking Society of Nephrology, Dialysis and Transplantation from MSD.

Acknowledgments

CKD-REIN is funded by the *Agence Nationale de la Recherche* through the 2010 "*Cohortes-Investissements d'Avenir*" program (ANR) and by the 2010 national *Programme Hospitalier de Recherche Clinique*. CKD-REIN is also supported through a public-private partnership with Amgen, Fresenius Medical Care, and GlaxoSmithKline (GSK), since 2012, Lilly France since 2013, and Otsuka Pharmaceutical since 2015, Baxter and Merck Sharp & Dohme-Chibret (MSD France) from 2012 to 2017, Sanofi-Genzyme from 2012 to 2015, Vifor Fresenius, and AstraZeneca, since 2018. Inserm Transfert set up and has managed this partnership since 2011. A specific project on drug optimization in CKD patients has been funded by the French National Agency for Medicines and Health Products Safety (ANSM).

The authors would like to thank the CKD-REIN study coordination staff for their efforts in setting up the CKD-REIN cohort: M.M., Elodie Speyer, Céline Lange, Sophie Renault, Reine

Ketchemin, Natalia Alencar de Pinho and all the clinical research associates. We thank Jo Ann Cahn for editing the English version.

The CKD-REIN Study Group. Steering committee and coordinators include: Carole Ayav, Serge Briançon, Dorothée Cannet, C.C., D.F., L.F., Yves-Edouard Herpe, C.J., M.L., Z.A.M., Christophe Pascal, B.M.R., B.S., Céline Lange, Karine Legrand, S.L., M.M. and Elodie Speyer.

CKD-REIN investigators/collaborators include: Thierry Hannedouche, Bruno Moulin, Sébastien Mailliez, Gaétan Lebrun, Eric Magnant, Gabriel Choukroun, Benjamin Deroure, Adeline Lacraz, Guy Lambrey, Jean Philippe, Bourdenx, Marie Essig, Thierry Lobbedez, Raymond Azar, Hacène Sekhri, Mustafa Smati, Mohamed Jamali, Alexandre Klein, Michel Delahousse, C.C., Séverine Martin, Isabelle Landru, Eric Thervet, Z.A.M., Philippe Lang, Xavier Belenfant, Pablo Urena, Carlos Vela, L.F., Dominique Chauveau, Viktor Panescu, Christian Noel, François Glowacki, Maxime Hoffmann, Maryvonne Hourmant, Dominique Besnier, Angelo Testa, François Kuentz, Philippe Zaoui, Charles Chazot, Laurent Juillard, Stéphane Burtey, Adrien Keller, Nassim Kamar, D.F. and M.L.

Supplemental material Table of Contents:

Supplementary material 1: Algorithms to assess causation and preventability

Supplementary material 2: Baseline characteristics of participants in the Chronic Kidney Disease-Renal Epidemiology and Information Network (before imputation)

Supplementary material 3: Details of types of adverse drug reactions

Supplementary material 4: Descriptions of adverse drug reactions according to the last estimated glomerular filtration rate reported before the reaction

Supplementary material 5: Distribution of serious adverse drug reactions causing or resulting from hospitalization according to the last estimated glomerular filtration rate reported before the reaction (n=145).

Supplementary material 6: Details of imputed drugs responsible for adverse drug reactions and of type of adverse drug reaction for the 5 pharmacological classes most frequently imputed.

Supplementary material 7: Sensitivity analyses by the Prentice, Williams, and Peterson (PWP) gap-time recurrent event time-to-event analysis

References:

1. Lazarou J, Pomeranz BH, Corey PN: Incidence of adverse drug reactions in hospitalized patients: a meta-analysis of prospective studies. *JAMA* 279: 1200–1205, 1998

2. Pouyanne P, Haramburu F, Imbs JL, Bégaud B: Admissions to hospital caused by adverse drug reactions: cross sectional incidence study. *BMJ* 320: 1036, 2000

3. Kongkaew C, Noyce PR, Ashcroft DM: Hospital admissions associated with adverse drug reactions: a systematic review of prospective observational studies. *Ann Pharmacother* 42: 1017–1025, 2008

4. Taché SV, Sönnichsen A, Ashcroft DM: Prevalence of Adverse Drug Events in Ambulatory Care: A Systematic Review. *Ann Pharmacother* 45: 977–989, 2011

5. Bénard-Laribière A, Miremont-Salamé G, Pérault-Pochat M-C, Noize P, Haramburu F, the EMIR Study Group on behalf of the French network of pharmacovigilance centres: Incidence of hospital admissions due to adverse drug reactions in France: the EMIR study. *Fundam Clin Pharmacol* 29: 106–111, 2015

6. Edwards IR, Aronson JK: Adverse drug reactions: definitions, diagnosis, and management. *The Lancet* 356: 1255–1259, 2000

7. Chan SL, Ang X, Sani LL, Ng HY, Winther MD, Liu JJ, Brunham LR, Chan A: Prevalence and characteristics of adverse drug reactions at admission to hospital: a prospective observational study: Prevalence of ADR-related hospitalization in Singapore. *Br J Clin Pharmacol* 82: 1636–1646, 2016

8. Alexopoulou A, Dourakis SP, Mantzoukis D, Pitsariotis T, Kandyli A, Deutsch M, Archimandritis AJ: Adverse drug reactions as a cause of hospital admissions: A 6-month experience in a single center in Greece. *Eur J Intern Med* 19: 505–510, 2008

9. Onder G, Pedone C, Landi F, Cesari M, Vedova CD, Bernabei R, Gambassi G: Adverse Drug Reactions as Cause of Hospital Admissions: Results from the Italian Group of Pharmacoepidemiology in the Elderly (GIFA). *J Am Geriatr Soc* 50: 1962–1968, 2002

10. Pirmohamed M, James S, Meakin S, Green C, Scott AK, Walley TJ, Farrar K, Park BK, Breckenridge AM: Adverse drug reactions as cause of admission to hospital: prospective analysis of 18 820 patients. *BMJ* 329: 15–19, 2004

11. Corsonello A, Pedone C, Corica F, Mussi C, Carbonin P, Incalzi RA: Concealed Renal

Insufficiency and Adverse Drug Reactions in Elderly Hospitalized Patients. *Arch Intern Med* 165: 790–795, 2005

12. Danial M, Hassali MA, Ong LM, Khan AH: Survivability of hospitalized chronic kidney disease (CKD) patients with moderate to severe estimated glomerular filtration rate (eGFR) after experiencing adverse drug reactions (ADRs) in a public healthcare center: a retrospective 3 year study. *BMC Pharmacol Toxicol* 19: 52, 2018

13. Bouvy JC, De Bruin ML, Koopmanschap MA: Epidemiology of Adverse Drug Reactions in Europe: A Review of Recent Observational Studies. *Drug Saf* 38: 437–453, 2015

14. Fattinger K, Roos M, Vergères P, Holenstein C, Kind B, Masche U, Stocker DN, Braunschweig S, Kullak-Ublick GA, Galeazzi RL, Follath F, Gasser T, Meier PJ: Epidemiology of drug exposure and adverse drug reactions in two Swiss departments of internal medicine. *Br J Clin Pharmacol* 49: 158–167, 2000

15. Pierson-Marchandise M, Gras V, Moragny J, Micallef J, Gaboriau L, Picard S, Choukroun G, Masmoudi K, Liabeuf S: The drugs that mostly frequently induce acute kidney injury: a case – noncase study of a pharmacovigilance database. *Br J Clin Pharmacol* 83: 1341–1349, 2017

16. Levey AS, Coresh J: Chronic kidney disease. *The Lancet* 379: 165–180, 2012

17. Hassan Y, Al-Ramahi R, Abd Aziz N, Ghazali R: Drug use and dosing in chronic kidney disease. *Ann Acad Med Singapore* 38: 1095–1103, 2009

Nolin TD: A Synopsis of Clinical Pharmacokinetic Alterations in Advanced CKD.
 Semin Dial 28: 325–329, 2015

19. Laville SM, Metzger M, Stengel B, Jacquelinet C, Combe C, Fouque D, Laville M, Frimat L, Ayav C, Speyer E, Robinson BM, Massy ZA, Liabeuf S, Chronic Kidney Disease-Renal Epidemiology and Information Network (CKD-REIN) Study Collaborators: Evaluation of the adequacy of drug prescriptions in patients with chronic kidney disease: results from the CKD-REIN cohort. *Br J Clin Pharmacol* 84: 2811–2823, 2018

20. Ginsberg JS, Zhan M, Diamantidis CJ, Woods C, Chen J, Fink JC: Patient-Reported and Actionable Safety Events in CKD. *J Am Soc Nephrol* 25: 1564–1573, 2014

21. Baigent C, Landray MJ, Reith C, Emberson J, Wheeler DC, Tomson C, Wanner C, Krane V, Cass A, Craig J, Neal B, Jiang L, Hooi LS, Levin A, Agodoa L, Gaziano M, Kasiske

B, Walker R, Massy ZA, Feldt-Rasmussen B, Krairittichai U, Ophascharoensuk V, Fellström B, Holdaas H, Tesar V, Wiecek A, Grobbee D, Zeeuw D de, Grönhagen-Riska C, Dasgupta T, Lewis D, Herrington W, Mafham M, Majoni W, Wallendszus K, Grimm R, Pedersen T, Tobert J, Armitage J, Baxter A, Bray C, Chen Y, Chen Z, Hill M, Knott C, Parish S, Simpson D, Sleight P, Young A, Collins R: The effects of lowering LDL cholesterol with simvastatin plus ezetimibe in patients with chronic kidney disease (Study of Heart and Renal Protection): a randomised placebo-controlled trial. *The Lancet* 377: 2181–2192, 2011

22. Liabeuf S, Ryckelynck J-P, Esper NE, Ureña P, Combe C, Dussol B, Fouque D, Vanhille P, Frimat L, Thervet E, Mentaverri R, Prié D, Choukroun G: Randomized Clinical Trial of Sevelamer Carbonate on Serum Klotho and Fibroblast Growth Factor 23 in CKD. *Clin J Am Soc Nephrol* 12: 1930–1940, 2017

23. Saheb Sharif-Askari F, Syed Sulaiman SA, Saheb Sharif-Askari N, Al Sayed Hussain A: Development of an Adverse Drug Reaction Risk Assessment Score among Hospitalized Patients with Chronic Kidney Disease. *PLoS ONE* [Internet] 9: 2014 Available from: https://www.ncbi.nlm.nih.gov/pmc/articles/PMC3995995/ [cited 2018 Dec 21]

24. Pecoits-Filho R, Fliser D, Tu C, Zee J, Bieber B, Wong MMY, Port F, Combe C, Lopes AA, Reichel H, Narita I, Stengel B, Robinson BM, Massy Z, CKDopps Investigators, Duttlinger C, Duttlinger J, Lonnemann G, Wada T, Yamagata K, Pisoni R, Calice da Silva V, Sesso R, Speyer E, Asahi K, Hoshino J, Perlman R, Sukul N, Young E: Prescription of renin-angiotensinaldosterone system inhibitors (RAASi) and its determinants in patients with advanced CKD under nephrologist care. *J Clin Hypertens* 21: 991–1001, 2019

25. Stengel B, Metzger M, Combe C, Jacquelinet C, Briançon S, Ayav C, Fouque D, Laville M, Frimat L, Pascal C, Herpe Y-É, Morel P, Deleuze J-F, Schanstra J, Lange C, Legrand K, Speyer E, Liabeuf S, Robinson BM, Massy ZA: Risk profile, quality of life and care of patients with moderate and advanced CKD. The French Chronic Kidney Disease – Renal Epidemiology and Information Network (CKD-REIN) Cohort Study. *Nephrol Dial Transplant* 2018

26. Girerd X, Radauceanu A, Achard JM, Fourcade J, Tournier B, Brillet G, Silhol F, Hanon
O: [Evaluation of patient compliance among hypertensive patients treated by specialists]. *Arch Mal Coeur Vaiss* 94: 839–842, 2001

27. Levey AS, Stevens LA, Schmid CH, Zhang Y (Lucy), Castro AF, Feldman HI, Kusek JW, Eggers P, Van Lente F, Greene T, Coresh J, for the CKD-EPI (Chronic Kidney Disease

Epidemiology Collaboration): A New Equation to Estimate Glomerular Filtration Rate. *Ann Intern Med* 150: 604, 2009

28. World Health Organization Collaborating Centre for Drug Statistics Methodology,
Oslo: Guidelines for ATC classification and DDD assignment 2016, 20th Edition. [Internet].
2016 Available from: www.whocc.no/atc_ddd_index/

29. Couchoud C, Stengel B, Landais P, Aldigier J-C, de Cornelissen F, Dabot C, Maheut H, Joyeux V, Kessler M, Labeeuw M, Isnard H, Jacquelinet C: The renal epidemiology and information network (REIN): a new registry for end-stage renal disease in France. *Nephrol Dial Transplant Off Publ Eur Dial Transpl Assoc - Eur Ren Assoc* 21: 411–418, 2006

30. World Health Organization, Quality Assurance and Safety of Medicines Team: Safety of medicines : a guide to detecting and reporting adverse drug reactions : why health professionals need to take action. [Internet]. 2002 Available from: https://apps.who.int/iris/bitstream/handle/10665/67378/WHO_EDM_QSM_2002.2.pdf?seque nce=1&isAllowed=y

31. Bégaud B, Evreux JC, Jouglard J, Lagier G: Imputation of the unexpected or toxic effects of drugs. Actualization of the method used in France. *Therapie* 40: 111–118, 1985

32. Naranjo CA, Busto U, Sellers EM, Sandor P, Ruiz I, Roberts EA, Janecek E, Domecq C, Greenblatt DJ: A method for estimating the probability of adverse drug reactions. *Clin Pharmacol Ther* 30: 239–245, 1981

33. Olivier P, Caron J, Haramburu F, Imbs J-L, Jonville-Béra A-P, Lagier G, Sgro C, Vial T, Montastruc J-L, Lapeyre-Mestre M: Validation d'une échelle de mesure : exemple de l'échelle française d'évitabilité des effets indésirables médicamenteux. *Thérapie* 60: 39–45, 2005

34. Prentice RL, Williams BJ, Peterson AV: On the regression analysis of multivariate failure time data. *Biometrika* 68: 373–379, 1981

35. Yang W, Jepson C, Xie D, Roy JA, Shou H, Hsu JY, Anderson AH, Landis JR, He J, Feldman HI, Investigators on behalf of the CRIC (CRIC) S: Statistical Methods for Recurrent Event Analysis in Cohort Studies of CKD. *Clin J Am Soc Nephrol* 12: 2066–2073, 2017

36. Van Buuren S: Multiple imputation of discrete and continuous data by fully conditional specification. *Stat Methods Med Res* 16: 219–242, 2007

37. Whittaker CF, Miklich MA, Patel RS, Fink JC: Medication Safety Principles and Practice in CKD. *Clin J Am Soc Nephrol* CJN.00580118, 2018

38. Tonelli M, Wiebe N, Manns BJ, Klarenbach SW, James MT, Ravani P, Pannu N, Himmelfarb J, Hemmelgarn BR: Comparison of the Complexity of Patients Seen by Different Medical Subspecialists in a Universal Health Care System. *JAMA Netw Open* 1: e184852–e184852, 2018

39. Schmidt IM, Hübner S, Nadal J, Titze S, Schmid M, Bärthlein B, Schlieper G, Dienemann T, Schultheiss UT, Meiselbach H, Köttgen A, Flöge J, Busch M, Kreutz R, Kielstein JT, Eckardt K-U: Patterns of medication use and the burden of polypharmacy in patients with chronic kidney disease: the German Chronic Kidney Disease study. *Clin Kidney J* 12: 663–672, 2019

40. Schneider MP, Hübner S, Titze SI, Schmid M, Nadal J, Schlieper G, Busch M, Baid-Agrawal S, Krane V, Wanner C, Kronenberg F, Eckardt K-U, on behalf of the GCKD Study Investigators: Implementation of the KDIGO guideline on lipid management requires a substantial increase in statin prescription rates. *Kidney Int* 88: 1411–1418, 2015

41. Massy ZA, Ferrières J, Bruckert E, Lange C, Liabeuf S, Velkovski-Rouyer M, Stengel
B: Achievement of Low-Density Lipoprotein Cholesterol Targets in CKD. *Kidney Int Rep*[Internet] Available from: https://doi.org/10.1016/j.ekir.2019.07.014 [cited 2019 Sep 16]

42. Fried TR, O'Leary J, Towle V, Goldstein MK, Trentalange M, Martin DK: Health Outcomes Associated with Polypharmacy in Community-Dwelling Older Adults: A Systematic Review. *J Am Geriatr Soc* 62: 2261–2272, 2014

43. Brookhart MA, Patrick AR, Dormuth C, Avorn J, Shrank W, Cadarette SM, Solomon DH: Adherence to lipid-lowering therapy and the use of preventive health services: an investigation of the healthy user effect. *Am J Epidemiol* 166: 348–354, 2007

44. Corsonello A, Pedone C, Lattanzio F, Onder G, Antonelli Incalzi R, Gruppo Italiano di Farmacovigilanza nell'Anziano (GIFA): Association between glomerular filtration rate and adverse drug reactions in elderly hospitalized patients: the role of the estimating equation. *Drugs Aging* 28: 379–390, 2011

45. Fawzy AM, Lip GYH: Pharmacokinetics and pharmacodynamics of oral anticoagulants used in atrial fibrillation. *Expert Opin Drug Metab Toxicol* 15: 381–398, 2019

46. Liabeuf S, Neirynck N, Drücke TB, Vanholder R, Massy ZA: Clinical Studies and Chronic Kidney Disease: What Did we Learn Recently? *Semin Nephrol* 34: 164–179, 2014

47. Shivanna S, Kolandaivelu K, Shashar M, Belghasim M, Al-Rabadi L, Balcells M, Zhang A, Weinberg J, Francis J, Pollastri MP, Edelman ER, Sherr DH, Chitalia VC: The Aryl Hydrocarbon Receptor is a Critical Regulator of Tissue Factor Stability and an Antithrombotic Target in Uremia. *J Am Soc Nephrol JASN* 27: 189–201, 2016

48. Hazell L, Shakir SAW: Under-Reporting of Adverse Drug Reactions. *Drug Saf* 29: 385–396, 2006

49. Agbabiaka TB, Savović J, Ernst E: Methods for Causality Assessment of Adverse Drug Reactions. *Drug Saf* 31: 21–37, 2008

50. Hutchinson TA, Flegel KM, HoPingKong H, Bloom WS, Kramer MS, Trummer EG: Reasons for disagreement in the standardized assessment of suspected adverse drug reactions. *Clin Pharmacol Ther* 34: 421–426, 1983

Figure legends

Figure 1: Description of the process for identifying and validating adverse drug reactions in

the CKD-REIN cohort

Figure 2: Incidence rates of adverse drug reactions and serious adverse drug reactions according to baseline estimated glomerular filtration rate.

Footnote: Incidence rates are represented with their 95% confidence interval whiskers. *P*-values test the difference between incidence rates according to baseline eGFR.

Figure 3: Distribution of serious adverse drug reactions causing or resulting from hospitalization according to baseline estimated glomerular filtration rate (n=145).

Footnote: Results are expressed as %. The denominators used are the total number of ADRs in patients with eGFR $< or \ge 30 \text{ mL/min}/1.73\text{m}^2$.

Figure 4: The most common pharmacological classes responsible for adverse drug reactions and serious adverse drug reactions according to baseline estimated glomerular filtration rate (n=751). Adverse drug reactions caused by agents acting on the renin-angiotensin system (RAS), antithrombotic agents, or diuretics accounted for 39% of adverse drug reactions. Serious adverse drug reactions caused by these three classes accounted for 58% of serious adverse drug reactions.

Footnote:

Results are expressed as %. The denominators used are the total number of ADRs in patients with eGFR < or \geq 30 mL/min/1.73m^2.

Table 1: Baseline characteristics of participants in the Chronic Kidney Disease-Renal

 Epidemiology and Information Network

	All (n=3033)	≥ 30 (n=1670)	<30 (n=1363)	Participants with missing data** (n=3033)
Age (years)	69 [60 - 76]	68 [59 - 75]	70 [61 - 78]	0%
<60	716 (24%)	421 (25%)	295 (22%)	
60 to 75	1400 (46%)	803 (48%)	597 (44%)	
≥ 75	917 (30%)	446 (27%)	471 (35%)	
Men	1982 (65%)	1121 (67%)	861 (63%)	0%
High school diploma or higher	1094 (36%)	648 (39%)	446 (33%)	1.7%
BMI	28 [25 - 32]	28 [25 - 32]	28 [25 - 32]	2.1%
$\geq 30 \ kg/m^2$	1075 (35%)	573 (34%)	502 (37%)	
Serum Albumin	4.0 [3.8 - 4.3]	4.1 [3.8 - 4.3]	4.0 [3.7 - 4.3]	18.9%
<3.5 g/L	293 (10%)	138 (8%)	155 (11%)	
UACR				11.2%
<30 mg/g	846 (28%)	625 (37%)	221 (16%)	
30 - 300 mg/g	955 (31%)	540 (32%)	415 (31%)	
>300 mg/g	1232 (41%)	505 (31%)	727 (53%)	
Anemia*	1236 (41%)	490 (29%)	746 (55%)	0.9%
Smoking status				0.8%
Smoker	360 (12%)	197 (12%)	163 (12%)	
Non smoker	1252 (41%)	701 (42%)	551 (40%)	
Ex-smoker	1421 (47%)	773(46%)	648 (48%)	
Diabetes	1304 (43%)	702 (42%)	602 (44%)	0.2%
AKI history	710 (23%)	350 (21%)	360 (26%)	8.0%
Cardiovascular history	1611 (53%)	854 (51%)	757 (56%)	1.4%
Hypertension	2751 (91%)	1493 (89%)	1258 (92%)	0.2%
Dyslipidemia	2229 (73%)	1216 (73%)	1011 (74%)	0.5%

Number of drugs	8 [5 - 10]	7 [5 - 10]	8 [6 - 11]	0.6%
< 5 drugs	593 (19%)	412 (25%)	181 (13%)	
5 to 10 drugs	1694 (56%)	915 (54%)	779 (57%)	
> 10 drugs	747 (25%)	344 (21%)	403 (30%)	
Poor adherence to medications	1888 (62%)	1010 (60%)	878 (64%)	1.0%

AKI: Acute kidney injury, BMI: Body mass index, eGFR: estimated glomerular filtration rate, UACR: Urine albumin to creatinine ratio

Median (interquartile range, IQR) or n (%)

*Anemia is defined by the 1968 WHO definition [*World Health Organization. Nutritional anaemias. Report of a WHO scientific group. Geneva, World Health Organization; 1968*]: <12 g/dL for women and <13 g/dL for men. **Missing data were imputed as specified in the methods section.

Type of ADRs	All ADRs	ADRs in patients with eGFR ≥30	ADRs in patients with eGFR <30 (n= 420)	
	(n=751)	(n=331)		
Renal and urinary disorders	150 (20%)	62 (19%)	88 (21%)	
Acute kidney injury	102	41	61	
Increased serum creatinine	40	20	20	
Other type of renal and urinary disorders	8	1	7	
Gastrointestinal disorders	119 (16%)	61 (18%)	58 (14%)	
Diarrhea	57	35	22	
Gastrointestinal conditions	24	8	16	
Other type of gastrointestinal disorders	38	18	20	
Musculoskeletal and connective tissue disorders	68 (9%)	34 (10%)	34 (8%)	
Contractures	35	21	14	
Muscle pain	22	10	12	
Other type of musculoskeletal and connective tissue	11	2	8	
disorders	11	3	o	
Hemorrhages and bleeding	67 (9%)	18 (5%)	49 (12%)	
Hemorrhages	34	8	26	
Hematoma	19	7	12	
Other type of hemorrhages and bleeding	14	3	11	
General disorders and administration site conditions	58 (8%)	26 (8%)	32 (8%)	
Peripheral edema	30	11	19	
Drug intolerance	8	3	5	
Other type of general disorders and administration	20	12	0	
site conditions	20	12	8	
Other type of ADRs	289 (38%)	130 (39%)	159 (38%)	

Table 2: Description of adverse drug reactions according to baseline estimated glomerular filtration rate.

ADRs: adverse drug reactions, eGFR: estimated glomerular filtration rate expressed in mL/min/1.73m² Results are expressed as n (%). The denominator used in column 1 is the total number of ADRs, and in columns 2 and 3, those of the ADRs in patients with eGFR < vs \geq 30 respectively.

P-value=0.025 tests the difference in the distribution of ADR type according to patient baseline eGFR.

	N(%) with	Incidence rate*	Unadjusted model		Adjusted model		
	ADRs (n=536)	IR [95% CI]	HR [95% CI]	<i>P</i> -value	HR [95% CI]	P-value	
Age (years)				0.10		0.10	
<60	113 (16)	13.2 [9.9; 17.6]	Reference		Reference		
60 to 75	271 (19)	15.2 [12.5; 18.5]	1.25 [1.00; 1.56]		1.06 [0.84; 1.34]		
≥ 75	152 (17)	14.0 [10.9; 18.0]	1.08 [0.85; 1.38]		0.85 [0.65; 1.11]		
Sex				0.08		0.04	
Men	332 (17)	13.2 [11.1; 15.7]					
Women	204 (19)	16.6 [13.4; 20.6]	1.17 [0.98; 1.39]		1.21 [1.01; 1.45]		
Educational level				0.09		0.99	
High	179 (16)	13.0 [10.2; 16.5]	Reference		Reference		
Low	357 (18)	15.2 [12.8; 17.9]	1.17 [0.97; 1.40]		1.00 [0.83; 1.21]		
eGFR				< 0.001		< 0.001	
\geq 30 mL/min per 1.73m ²	244 (15)	10.8 [8.9; 13.2]	Reference		Reference		
< 30 mL/min per 1.73m ²	292 (21)	19.4 [16.2; 23.2]	1.74 [1.47; 2.07]		1.56 [1.30; 1.87]		
Serum Albumin				0.06		0.31	
$\geq 3.5 \ g/dL$	476 (17)	13.9 [12.0; 16.0]	Reference		Reference		
<3.5 g/dL	60 (20)	19.7 [12.9; 29.9]	1.38 [1.00; 1.90]		1.19 [0.85; 1.65]		
UACR				0.06		0.90	
<30 mg/g	139 (16)	13.1 [10.0; 17.1]	Reference		Reference		
30 – 300 mg/g	165 (17)	12.9 [10.0; 16.7]	1.09 [0.85; 1.39]		0.95 [0.75; 1.22]		
>300 mg/g	232 (19)	16.6 [13.5; 20.4]	1.29 [1.03; 1.62]		1.00 [0.79; 1.27]		
Anemia**				< 0.001		0.06	
Without anemia	287 (16)	12.3 [10.3; 14.8]	Reference		Reference		
With anemia	249 (20)	17.7 [14.6; 21.5]	1.45 [1.23; 1.72]		1.19 [0.99; 1.42]		
Diabetes				0.01		0.78	
Without diabetes	274 (16)	12.1 [10.0; 14.7]	Reference		Reference		

Table 3: Associations of participant characteristics with adverse drug reactions

With diabetes	262 (20)	17.4 [14.4; 21.0]	1.31 [1.10; 1.55]		1.03 [0.85; 1.24]	
AKI history				< 0.001		0.01
Without AKI history	379 (16)	12.9 [11.0; 15.2]	Reference		Reference	
With AKI history	157 (22)	19.4 [15.0; 25.1]	1.48 [1.21; 1.81]		1.32 [1.08; 1.61]	
Cardiovascular history						
Without cardiovascular history	215 (15)	11.3 [9.1; 14.0]	Reference		Reference	
With cardiovascular history	321 (20)	17.2 [14.6; 20.4]	1.42 [1.19; 1.69]	< 0.001	1.24 [1.02; 1.50]	0.03
Hypertension				0.06		0.71
Without hypertension	38 (13)	9.4 [5.4; 16.1]	Reference		Reference	
With hypertension	498 (18)	14.9 [13.0; 17.1]	1.37 [0.99; 1.91]		1.07 [0.76; 1.51]	
Baseline number of drugs/patient				< 0.001		0.01
< 5 drugs	62 (10)	7.5 [5.0; 11.2]	Reference		Reference	
5 to10	301 (18)	13.9 [11.6; 16.7]	1.86 [1.41; 2.44]		1.50 [1.12; 2.02]	
> 10	173 (9)	21.7 [17.3; 27.3]	2.59 [1.93; 3.46]		1.71 [1.21; 2.41]	
Adherence to medication				< 0.001		0.01
Good	158 (14)	10.7 [8.3; 13.7]	Reference		Reference	
Poor	378 (20)	16.7 [14.3; 19.6]	1.52 [1.26; 1.83]		1.36 [1.12; 1.64]	

AKI: Acute kidney injury, CI: Confidence interval, eGFR: estimated Glomerular filtration rate, HR: hazard ratio, IR: incidence rate, UACR: Urine albumin to creatinine ratio

P-values used Wald Chi-square test for global variable effect.

*Incidence rates are expressed per 100 person-years.

**Anemia is defined by the 1968 WHO definition [*World Health Organization. Nutritional anaemias. Report of a WHO scientific group. Geneva, World Health Organization; 1968*]: <12 g/dL for women and <13 g/dL for men.

	N(%) with	Incidence rate*	Unadjusted model		Adjusted model		
	ADRs (n=125)	IR [95% CI]	HR [95% CI]	P-value	HR [95% CI]	P-value	
Age (years)				0.15		0.32	
<60	26 (4)	2.4 [0.9; 6.9]	Reference		Reference		
60 to 75	52 (4)	2.4 [1.1; 5.1]	1.02 [0.64; 1.63]		0.71 [0.44; 1.17]		
≥ 75	47 (5)	3.4 [1.6; 7.5]	1.45 [0.90; 2.34]		0.91 [0.54; 1.51]		
Sex				0.91		0.74	
Men	82 (4)	2.7 [1.5; 4.8]	Reference		Reference		
Women	43 (4)	2.8 [1.3; 6.0]	0.98 [0.68; 1.42]		1.06 [0.73; 1.55]		
eGFR				< 0.001		0.01	
\geq 30 mL/min per 1.73m ²	49 (3)	1.8 [0.9; 3.4]	Reference		Reference		
< 30 mL/min per 1.73m ²	76 (6)	4.0 [2.4; 6.8]	2.17 [1.51; 3.11]		1.82 [1.25; 2.63]		
BMI				0.05		0.46	
<30 kg/m ²	70 (4)	2.4 [1.3; 4.5]	Reference		Reference		
≥30 kg/m²	55 (5)	3.3 [1.6; 6.8]	1.42 [0.99; 2.04]		1.16 [0.79; 1.70]		
Anemia**				0.001		0.15	
Without anemia	60 (3)	2.0 [1.0; 4.1]	Reference		Reference		
With anemia	65 (5)	3.9 [2.0; 7.5]	1.77 [1.25; 2.52]		1.31 [0.91; 1.89]		
Diabetes				0.03		0.89	
Without diabetes	60 (3)	2.3 [1.2; 4.3]	Reference		Reference		
With diabetes	65 (5)	3.3 [1.8; 6.1]	1.47 [1.03; 2.09]		0.97 [0.65; 1.44]		
AKI history				0.01		0.12	
Without AKI history	84 (4)	2.3 [1.3; 4.2]	Reference		Reference		
With AKI history	41 (6)	4.0 [1.8; 9.3]	1.67 [1.12; 2.49]		1.37 [0.92; 2.05]		
Cardiovascular history				< 0.001		0.01	
Without cardiovascular history	35 (2)	1.5 [0.7; 3.4]	Reference		Reference		
With cardiovascular history	90 (6)	3.9 [2.4; 6.3]	2.45 [1.65; 3.63]		1.93 [1.25; 2.98]		

Table 4: Associations of participant characteristics with serious adverse drug reactions

Baseline number of drugs/ patients				< 0.001		0.05
< 5	10(2)	1.1 [0.2; 4.6]	Reference		Reference	
5 to10	62 (4)	2.4 [1.3; 4.3]	2.29 [1.17; 4.46]		1.53 [0.76; 3.10]	
> 10	53 (3)	5.0 [2.7; 9.4]	4.70 [2.39; 9.23]		2.31 [1.07; 5.01]	
Adherence to medication				0.01		0.03
Good	30(3)	1.8 [0.8; 4.3]	Reference		Reference	
Poor	95 (5)	3.3 [2.0; 5.4]	1.95 [1.29; 2.94]		1.59 [1.05; 2.42]	

AKI: Acute kidney injury, BMI: body mass index, CI: Confidence interval, eGFR: estimated Glomerular filtration rate, HR: hazard ratio, IR: incidence rate

P-values used Wald Chi-square test for global variable effect.

*Incidence rates are expressed per 100 person-years.

**Anemia is defined by the 1968 WHO definition [*World Health Organization. Nutritional anaemias. Report of a WHO scientific group. Geneva, World Health Organization; 1968*]: <12 g/dL for women and <13 g/dL for men.